Author’s response to reviews

Title: Atypical antipsychotics induce human osteoblasts apoptosis via Wnt/β-catenin signaling

Authors:
Peifan Li (232550646@qq.com)
yiming wang (754603457@qq.com)
xingde liu (2360040895@qq.com)
zhen zhou (623219198@qq.com)
Jun Wang (wangjunjessie@163.com)
Haiyan Zhou (402184255@qq.com)
Lei Zheng (756203338@qq.com)
Lixia Yang (980657990@qq.com)
Pingxia Xie (495984837@qq.com)
Da Zhou (1014677427@qq.com)

Version: 2 Date: 11 Jan 2019

Author’s response to reviews:

PHAT-D-18-00073R1

Atypical antipsychotics induce human osteoblasts apoptosis via Wnt/β-catenin signaling

Peifan Li; yiming wang; xingde liu; zhen zhou; Jun Wang; Haiyan Zhou; Lei Zheng; Lixia Yang

Dear Editor:

Thanks for your proposal and I have carried out some essential revisions according to your request.
Yours sincerely

Peifan Li

• Change in authorship

All authors agree with all proposed changes in authorship of submitted manuscripts, and completed change of authorship form.

• Funding

These funding body do not participate in data collection, analysis, and interpretation of data and so on, but provide financial support. We have revised it.

• Image attribution

The images depicted in figure 5 are my own.

• Declarations

I have add a heading Declarations after the Abbreviations section.

• Clean manuscript

I have upload my manuscript as a single, final, clean version.